Author’s response to reviews

Title: Association of matrix metalloprotease 1, 3, and 12 polymorphisms with rheumatic heart disease in a Chinese Han population

Authors:

Wei Hu (william200603@sina.com)
Yujia Ye (yeyujia626@163.com)
Yirui Yin (tom.edward@foxmail.com)
Peng Sang (speng431@163.com)
Linhua Li (llhdoctor@163.com)
Jing Wang (wangjingmd@gmail.com)
Wen Wan (wanwen_wanwen@sina.com)
Rui Li (roddicli@sina.com)
Xiangfeng Bai (Baixiangfeng2015@163.com)
Yuehui Xie (xiyuehuixyh@sina.com)
Zhaohui Meng (mengzhgroup@aliyun.com)

Version: 3 Date: 25 Jan 2018

Author’s response to reviews:

Dear Matteo Pasini,

We greatly appreciate for your warm work earnestly.

We have revised our paper according to your advice, we hope this revision can make our paper more acceptable. Some of your questions were answered below.

Thank you and all the reviewers for the kind advice.

Sincerely yours,
Zhaohui Meng

Responds:

1. Please include the email addresses for all authors on the title page. The corresponding author should still be indicated.

Answer: We have added the email addresses for all authors on the title page. The corresponding authors were marked as “*”.

2. About data availability statements.

Answer: The datasets used and/or analysed during the current study available from the corresponding author on reasonable request. The email address: mengzhgroup@aliyun.com.

3. Please include the ethics committee’s reference number if appropriate.

Answer: We are very sorry that our medical ethics committee has not provided us with the ethics committee’s reference number.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Answer: We have uploaded our manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables were clean versions.